Author’s response to reviews

Title: Recurrence of visceral and muco-cutaneous leishmaniasis in a patient under immunosuppressive therapy

Authors:

Gilles Darcis (gdarcis@chu.ulg.ac.be)
Gert Van der Auwera (GVdAuwera@itg.be)
Jean-Baptiste Giot (jbgiot@chu.ulg.ac.be)
Marie-Pierre Hayette (mphayette@chu.ulg.ac.be)
Françoise Tassin (francoise.tassin@chu.ulg.ac.be)
Jorge Arrese Estrada (j.arrese@ulg.ac.be)
Lieselotte Cnops (lcnops@itg.be)
Michel Moutschen (michel.moutschen@ulg.ac.be)
Laurence De Leval (Laurence.DeLeval@chuv.ch)
Philippe Leonard (philippe.leonard@chu.ulg.ac.be)

Version: 2 Date: 07 Jun 2017

Author’s response to reviews:

We warmly thank the editor for the second review of our manuscript entitled: "Recurrence of visceral and muco-cutaneous leishmaniasis in a patient under immunosuppressive therapy ", which was again helpful. You will find enclosed a revised version of our manuscript. We have incorporated all recommended changes.

We hope that you will find it suitable for publication in BMC infectious diseases and look forward to hearing from you.